Author's response to reviews

Title: Rationale, design and implementation protocol of an electronic health record Integrated Clinical Prediction Rule (iCPR) randomized trial in primary care

Authors:

Devin M Mann (dmann@bu.edu)
Joseph L Kannry (joseph.kannry@mssm.edu)
Daniel Edonyabo (daniel.edonyabo@mssm.edu)
Alice C Li (alicecli@gmail.com)
Jacqueline Arciniega (jacqueline.arciniega@nyumc.org)
James Stulman (james.stulman@mssm.edu)
Lucas Romero (lucas.romero@nyumc.org)
Juan Wisnivesky (juan.wisnivesky@mssm.edu)
Rhodes Adler (rhodes.adler@mssm.edu)
Thomas P McGinn (tmcginn@nshs.edu)

Version: 2 Date: 22 July 2011

Author's response to reviews:

July 22, 2011

Dear Editor,

Attached is a revised version of the manuscript now entitled “Rationale, design and implementation protocol of an electronic health record Integrated Clinical Prediction Rule (iCPR) randomized trial in primary care” which we ask to be considered for publication in Implementation Science as an original study protocol.

As requested we have
1. Revised the formatting to meet the IS requirements
2. Added the CONSORT checklist and ensured the manuscript conforms to CONSORT guidelines for RCT’s.
3. Added the missing sections of competing interests and author contributions

If there are any other changes required please let me know.

Sincerely,

Devin M. Mann, MD, MS P: 617-638-8021
Assistant Professor of Medicine
Boston University School of Medicine